

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Fontaine, Francois; Nyborg Jensen, Janne; Vejlin, Rune Majlund

## Working Paper Wealth, Portfolios, and Unemployment Duration

IZA Discussion Papers, No. 13423

**Provided in Cooperation with:** IZA – Institute of Labor Economics

*Suggested Citation:* Fontaine, Francois; Nyborg Jensen, Janne; Vejlin, Rune Majlund (2020) : Wealth, Portfolios, and Unemployment Duration, IZA Discussion Papers, No. 13423, Institute of Labor Economics (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/223865

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



## WWW.ECONSTOR.EU



Initiated by Deutsche Post Foundation

# DISCUSSION PAPER SERIES

IZA DP No. 13423

Wealth, Portfolios, and Unemployment Duration

François Fontaine Janne Nyborg Jensen Rune Vejlin

JUNE 2020



Initiated by Deutsche Post Foundation

## DISCUSSION PAPER SERIES

IZA DP No. 13423

# Wealth, Portfolios, and Unemployment Duration

#### **François Fontaine** *PSE, Universite Paris Pantheon Sorbonne and IZA*

Janne Nyborg Jensen

Rune Vejlin Aarhus University and IZA

JUNE 2020

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

# ABSTRACT

# Wealth, Portfolios, and Unemployment Duration<sup>\*</sup>

We use administrative data on individual balance sheets in Denmark to document how an individual's financial position affects job search behavior. We look at the effect of wealth at the entry into unemployment on the exit rate from unemployment as well as the effect on the subsequent match quality. The detailed data allows us not only to distinguish between liquid and illiquid parts, but also to decompose each of them into assets and liabilities. The decomposition of wealth into these four components is key to understanding how wealth affects job finding rates. In particular, we show that liquid assets reduce the probability of becoming re-employed, but we do not see an effect of liquid liabilities or the illiquid wealth components, while interest payments speed up re-employment. The results on subsequent match quality in form of job duration and wages are mixed.

JEL Classification:	J6, J64, J65
Keywords:	non-employment duration, wealth composition
	job search behavior

#### Corresponding author:

Rune Vejlin Department of Economics and Business Aarhus University Fuglesangs Alle 4 8210 Aarhus V Denmark E-mail: rvejlin@econ.au.dk

<sup>\*</sup> We thank Jake Bradley, Nuno Coimbra, Axel Gottfries, Rasmus Lentz, Daniel le Maire, Simon Mongey, Roland Rathelot and Michael Svarer, as well as the participants of the Copenhagen Workshop on Mismatch and Matching 2019, the Nottingham CFCM seminar and the PSE internal macro workshop for comments. We would like to thank Jacob Jensen for excellent research assistance. We acknowledge financial support from the Danish Social Sciences Research Council (grant no. DFF - 6109-00037). All mistakes are our own.

## 1 Introduction

Understanding job search behavior of unemployed workers is key for many policy questions such as optimal unemployment insurance, workfare, hiring and firing policies, active labor market policies, etc. One component affecting job search behavior is self-insurance through savings.<sup>1</sup> In this paper, we show that, beyond the distinction between liquid and illiquid wealth, the decomposition of wealth between assets and liabilities is important to understand job finding rates. In particular, we find that only liquid assets affect job finding, while both illiquid assets and liabilities in general do not.

Decomposing net wealth by liquidity is not a new idea. The degree of liquidity of the workers' portfolios has indeed been shown to be important for a number of economic outcomes like consumption in general (Gross and Souleles, 2002) and the sensitivity of consumption to income shocks or the response to fiscal stimulus (Kaplan, Violante, and Weidner, 2014).<sup>2</sup> We confirm previous results showing that the distinction between liquid and illiquid wealth matters, but we go one step beyond distinguishing between different degrees of liquidity and show the importance of separating assets from liabilities in order to understand heterogeneity in job search behaviors.

It is often complicated to find data where labor market statuses and detailed individual financial information are provided. For that reason, evidence on the effect of wealth on the exit rate from unemployment is often focused on liquid wealth. For example, Card, Chetty, and Weber (2007) use a quasi-experiment based on a discontinuity in the eligibility to severance payments in Austria that creates liquidity shocks at the entry into unemployment and measures the effect of liquidity on the exit rate from unemployment. A transfer equivalent to two months of wages at the entry into unemployment is shown to reduce the exit rate by 10% around the discontinuity. Basten, Fagereng, and Telle (2014) use a similar setup by exploiting the same type of discontinuity in severance payments in Norway. They find that the severance payments increase average non-employment durations and lower the fraction re-employed after one year. The advantage of the approach used in the two papers, with respect to previous works, is that it solves the problem of endogeneity of the wealth variables by using a regression-discontinuity (RD) design. In our paper we cannot exploit such a discontinuity but, since we follow the same individuals over time, we rely on a fixed-effect approach combined with a very rich set of control variables.

In our detailed Danish population-wide data, we do not see any effect of total net wealth on the re-employment probability. However, this result hides important aspects of

 $<sup>^1\</sup>mathrm{Early}$  examples are Bloemen and Stancanelli, 2001 or Alexopoulos and Gladden, 2004.

<sup>&</sup>lt;sup>2</sup>There is a whole literature on the impact of housing, the most prevalent illiquid asset, on job search and job choices (e.g. Munch, Rosholm, and Svarer, 2008, Coulson and Fisher, 2009, Winkler, 2011, Head and Lloyd-Ellis, 2012, Caliendo, Gielen, and Mahlstedt, 2015 or le Maire and He, 2018). In most countries one cannot easily borrow against real estate holdings which reduce their value in terms of insurance. In that case, being a home owner could actually reduce mobility on the labor market by increasing the adjustment costs implied by a job change.

how the different wealth components affect the re-employment probability. Decomposing net wealth, by degree of liquidity first and then by separating assets and liabilities, we find a significant effect of liquid assets on the probability of becoming re-employed after 12 months. Moreover, the effect for liquid assets is more pronounced for those with low levels of liquid assets. What is new to this paper, in comparison with Card, Chetty, and Weber (2007) and Basten, Fagereng, and Telle (2014) who only look at liquid assets, is that we study the impact of liabilities *per se*, both liquid (such as credit card debt and bank debt) and illiquid (such as mortgages) liabilities. We show that neither illiquid assets nor illiquid liabilities affect the re-employment probability after controlling for individual fixed effects, and the results for liquid liabilities are mixed but qualitatively small. On the contrary, interest payments to banks speed up job finding.

In that way, we complement a literature that has shown that net wealth, which is assets minus liabilities, is the key variable linking saving decisions and job search activities (e.g. Algan, Cheron, Hairault, and Langot, 2003, Lentz and Tranas, 2005, Rendon, 2006, Lentz, 2009, Lise, 2013, or Shi and Chaumont, 2017). This literature gives a simple explanation for the existing empirical evidence: In an environment with credit constraints, precautionary savings allow smoothing of consumption in case of unemployment shocks and thus affect the intensity of search effort and potentially the reservation job quality. These theoretical papers use net wealth instead of separating assets and liabilities because in a canonical model of precautionary savings, it is not optimal to borrow and keep low return liquidity at the same time. At odds with this prediction, Telyukova, 2013 states that a little more than a quarter of all US households have both credit card debt and liquidity at the same time, and, in the same way, we find in our data that 71% of the workers hold both liquid assets and liquid debt.<sup>3</sup>. Our paper provides new empirical evidence, in the job search context, that calls for understanding the impact of wealth heterogeneity on the cost of unemployment by separating assets and liabilities.

The paper proceeds as follows. Section 2 presents the data used for the estimations, and Section 3 outlines the empirical strategy. Section 4 analyzes the results and finally, Section 5 concludes.

### 2 Data

We use administrative register data from Denmark for the period 1997 to 2011 containing socio-demographic-, labor market-, and financial information. Our point of departure is the DREAM data set which records the primary public transfer to all individuals in Denmark each week. From this register we sample all individuals recorded as unemployed at least once. We use a narrow definition of unemployment and only sample those who

 $<sup>^{3}80\%</sup>$  of workers have both assets and debt at the same time, and almost everyone in the entire sample (99.6%) has either assets or debt. 71% hold both liquid assets and liquid debt, while 39% have illiquid assets as well as illiquid debt (44% have either illiquid assets or debt).

where not on other public benefits before the beginning of the current unemployment spell, i.e. those who are on unemployment benefit or social insurance and ready for the labor market.<sup>4</sup> We use this data set to later extract unemployment spells. We merge this data with daily job spell information from a matched employer-employee data set. Finally, these two data sources are combined with yearly socio-demographic information such as detailed information on wealth holdings, interest payments, hourly wages, public transfers, and background characteristics as well as household information. We use the public transfers together with unemployment spell data to create a variable for the unemployment benefit level, which is used as a key control variable.<sup>5</sup> We merge background information including financial information onto the unemployment spells using the information at the end of the year prior to the start of the unemployment spell.<sup>6</sup> The financial information is collected by the tax authorities, and there is very limited self-reporting. Most of the information is automatically reported by the banks, mortgage institutions etc. Thus, the data is considered to be of very high quality.

One of the main advantages of the data is the very detailed information on wealth. We know the portfolio composition of individual wealth and are able to split wealth into assets and liabilities for cash holdings, stocks and bond holdings as well as the cash value of properties owned by the individuals and the amount owed to credit institutions. We use this information to decompose assets and liabilities according to liquidity. The first measure is liquid assets which consist of bank deposits and stock and bond holdings. Second, we measure liquid liabilities as the amounts owed to credit institutions such as bank or credit card companies. The third measure we use is illiquid assets measuring the value of properties owned by the individuals.<sup>7</sup> Finally, illiquid liabilities measure bond debts primarily capture mortgage dept. We measure all these variables at the individual annual level. In the case where individuals live as a couple, we can separate the household wealth between the two and then control for the wealth of the partner in the estimation stage.<sup>8</sup>

This gives us a sample of 2,095,083 unemployment spells experienced by 1,175,438 individuals. Notice that 40.55% of individuals experience more than one unemployment

<sup>&</sup>lt;sup>4</sup>Our definition of unemployment includes workers on unemployment benefit and social insurance if they are deemed ready for the labor market. It consists of the following: people on unemployment benefits ('Dagpenge, ledighed'), social assistance and ready to take a job ('Arbejdsmarkedsparate kontanthjælpsbrugere'), transitional benefits ('Overgangsydelse'), or special unemployment benefits ('Ledighedsydelse').

<sup>&</sup>lt;sup>5</sup>Since UI benefits depend on previous wages, they are not the same for all workers although approximately 85 percent of workers receive the maximum level.

<sup>&</sup>lt;sup>6</sup>In the robustness section, we show that our main results hold if we condition on unemployment that starts within the first six months of the year.

<sup>&</sup>lt;sup>7</sup>The value of properties is based on public valuations which form the basis for property taxation. House and apartment values are updated every second year, and individuals can complain if they do not think that the valuation is fair. In general, the tax valuations are thought to be of good quality.

 $<sup>^{8}</sup>$ We defined couples as both married and cohabiting couples. Cohabitation is very common in Denmark. In some cases (such as shared bank accounts) the wealth is owned equally, while in other cases (such as owning a house) the ownership is more explicit with one partner owning x percent and the other 1-x percent.

spell, while 90.69% of the individuals experience at most three unemployment spells during our sample period. Tables 1, 2, and 3 show background information, labor market and financial information, and information about the outcomes for the sample used in the main analysis which consists of 1,398,727 observations.<sup>9</sup>

	Mean	SD
Personal characteristics		
Age	38.01	$11,\!50$
Female $(\%)$	38.60	48.68
Single $(\%)$	41.60	49.29
Married/cohabitating $(\%)$	58.40	49.29
Education (%)		
Less than high school	37.26	48.35
High school/vocational education	47.35	49.93
College or more	15.39	36.08
Occupation (%)		
Unknown	26.32	44.04
Military	0.62	7.85
Management at highest level	1.02	10.03
Job requiring knowledge at the highest level	5.74	23.25
Job req. knowledge at the medium level	7.58	26.48
Office jobs	6.60	24.84
Sales/Service/Care	8.88	28.46
Job in agriculture, forrestry, fishing req. basic level knowledge	1.25	11.12
Craftsmanlike jobs	16.86	37.44
Process tech., maschine techm., transportation, construction	11.06	31.36
Other	14.06	34.76
Number of observations	1,398	3,727

TABLE	1:	Descriptive	STATISTICS,	BACKGROUND	INFORMATION
-------	----	-------------	-------------	------------	-------------

Note: All variables are measured the year before the beginning of the unemployment spell.

In Table 1 it can be seen that almost 39% of our sample are female, 58% are married or cohabiting with a partner, and 42% are singles. The percentage of singles is a bit higher in our sample compared to all individuals in Denmark for the same time period, where 35% are single and 65% are married or cohabiting. Almost half of our sample have a high school or vocational education as their highest attained education, and 15% have completed college or higher.

Table 2 provides information about the labor market attachment and financial outcomes for the individuals. The mean duration of unemployment is 18 weeks. Workers on

 $<sup>^{9}</sup>$ The main sample used is those experiencing multiple spells and whose unemployment spell begins at least 12 months before the end of the sampling period.

average have 12 years of experience, and 75% are members of an unemployment fund. Average total income the year before the beginning of the unemployment spell is 225,000 DKK (34,500 USD) with 187,000 DKK (29,000 USD) coming from wage income.

	Mean	SD
Labor market attachment		
Duration of unemployment (weeks)	17.74	21.22
Member of an unemployment fond (A-kasse) (%)	74.83	43.40
Tenure in previous job (weeks)	48.44	68.96
Total experience	12.40	9.16
Experience gained, 1 year before U	0.72	0.33
Experience gained, 2 years before U	0.69	0.36
Experience gained, 3 years before U	0.66	0.38
Degree of unemployment, 1 year before U $(\%)$	12.09	18.70
Degree of unemployment, 2 years before U $(\%)$	12.35	20.59
Degree of unemployment, 3 years before U $(\%)$	12.96	21.89
Other financial variables (in year 2000 DKK)		
Total income	225,056	114,358
Wage income, 1 year before U	186,844	107,017
Wage income, 2 years before U	176,908	112,661
Wage income, 3 years before U	166,548	115,528
Hourly wage, 1 year before U	157	103
Hourly wage, 2 years before U	152	96
Hourly wage, 3 years before U	148	88
Number of observations	1,398	8,727

TABLE 2: DESCRIPTIVE STATISTICS, LABOR MARKET AND FINANCIAL INFORMATION

Note: All variables are measured the year before the beginning of the unemployment spell (except the duration of unemployment). All financial values are measured in 2000 DKK prices.

Our main outcome variable is the probability of being re-employed within the first 12 months after the unemployment spell begins, but we also check whether our results hold for different time horizons. As can be seen in Table 3, 82% of our sample become re-employed within the first 12 months after the unemployment spell begins.

Furthermore, we investigate the match quality of the job following an unemployment spell by looking at the probability that the first job after re-employment lasts at least one year, that the total employment spell after re-employment lasts at least two years, and also at the re-employment wage.<sup>10</sup> The probability that the first job after re-employment lasts at least one year is 31%, while the probability of the total employment spell lasting

 $<sup>^{10}{\</sup>rm Job}$  spell length refers to the length of the first job held after unemployment, while employment spell length refers to the length of the entire spell without breaks in employment. The employment spell can thus consist of several jobs.

	Mean	SD	Within SD
Fraction re-employed after $(\%)$			
6 months	69.42	46.08	
9 months	77.94	41.47	
12 months	82.42	38.06	
24 months	90.49	29.34	
36 months	93.32	25.15	
Match quality of job following unemployment			
First job lasts at least 1 year	31.16	46.31	
Total employment spell lasts at least 2 years	29.12	45.43	
Re-employment labor income (monthly, year 2000 DKK)	41,698	50,905	
Wealth variables (in year 2000 DKK)			
Net wealth	94.479	493.233	112.029
Liquid wealth	-31.799	281.051	67.675
Illiquid wealth	163.033	450.259	101.031
Liquid assets	57.278	223.175	32.611
Liquid liabilities	89.077	195.133	48.250
Illiquid assets	343.446	646.110	146.328
Illiquid liabilities	180.413	330.061	85.455
Positive interest payments, banks	1.078	6.021	,
Negative interest payments, banks	6.303	9.887	
Negative interest payments, mortgage institutions	10.238	16.508	
respected metrose paymentes, metrogage motioned	10,200	-0,000	
Number of observations	1,398	8,727	

TABLE 3: DESCRIPTIVE STATISTICS, WEALTH, AND OUTCOMES

Note: Wealth variables are measured the year before the beginning of the unemployment spell. All financial values are measured in 2000 DKK prices. The third column ("Within SD") refers to within-person variation.

at least two years is 29%.<sup>11</sup> Mean re-employment wages are equal to 41,700 DKK (6,400 USD) monthly.

The mean amounts of liquid assets and liabilities are 57,000 DKK (8,800 USD) and 89,000 DKK (13,700 USD) respectively. A lot of the individuals have low liquid net wealth which lowers the mean. For the illiquid wealth components, the mean amounts of illiquid assets and liabilities are 343,000 DKK (52,700 USD) and 180,000 DKK (27,600 USD), respectively. A lot of individuals do not have any illiquid net wealth, but those who do tend to hold large amounts. Recall that this primarily reflects homeownership. On average, individuals receive 1000 DKK (153 USD) in interest payments from banks but pay 6,000 DKK (900 USD) to banks and more than 10,000 DKK (1,500 USD) to credit institutions.

<sup>&</sup>lt;sup>11</sup>On average, the first job lasts a bit more than a year (58 weeks), while the total employment spell lasts almost two years (93 weeks).



FIGURE 1: DECILES OF WEALTH VARIABLES

Note: The figure shows the 10 deciles of the four wealth components separately. The top graph shows the deciles for liquid assets and liabilities, while the bottom graph shows the deciles for illiquid assets and liabilities.

Figure 1 gives more details about the distribution of the wealth variables. The first graph shows the 10 deciles of the liquid assets and liquid liabilities, and the second graph shows the same for the illiquid assets and liabilities. As can be seen from the figure, the distributions of the wealth variables are very different, as expected. Even the individuals in the first decile of the liquid asset distribution hold a positive amount (only 7.1% hold zero liquid assets), but the holdings of liquid assets are in general not large. A larger fraction has no liquid liabilities (22.6% have zero liquid liabilities), but the top 10% owe almost twice as much as the top 10% of individuals hold in liquid assets. For the illiquid components, a lot of individuals do not hold any positive or negative amounts, i.e. a lot of individuals are not home owners. 57.1% hold zero illiquid assets, while 60.1% have no mortgage debt. All wealth variables show considerable dispersion.

A main motivating question for this paper is whether job search behavior is affected

not just by total wealth, but by the portfolio of wealth. In a recent paper by Kaplan, Violante, and Weidner (2014) households are divided into hand-to-mouth (HTM), which are then split into wealthy HTM and poor HTM. Both groups do not have any net liquid wealth (liquid assets minus liquid liabilities), but only the wealthy HTM households hold positive amounts of illiquid net wealth. Empirically, it is not straightforward to measure the HTM-status of the individuals. If we follow the definitions by Kaplan, Violante, and Weidner (2014))<sup>12</sup>, 61% of the individuals in our sample are HTM. Of these, 56% are poor HTM (no illiquid net wealth), while 44% are wealthy HTM (positive illiquid wealth). This is a higher share of HTM individuals compared to Kaplan, Violante, and Weidner (2014)<sup>13</sup> comes from the fact that our financial variables are measured at the end of December a high-spending period, and from our selected sample that only includes those who are unemployed at some point in the observation period. Another reason for this difference could be that it is easier to obtain overdraft facilities in Denmark than in the US or other countries.

## 3 Empirical strategy

Several previous studies have investigated the question of how individual wealth affects job search behavior using proportional hazard models or other models assuming that all unobserved heterogeneity is uncorrelated with observable characteristics. If this is not the case, the estimates will not be consistent. If for example individuals differ in their risk aversion, this is likely to be correlated with their wealth holdings. In this case, the hazard model would not be able to control for such differences, since it only controls for unobserved heterogeneity that is uncorrelated with the independent variables.

We prefer to control for constant unobserved individual heterogeneity that is potentially correlated with the independent variables. To do this, we perform an individual fixed effects estimation. We consider the following equation:

$$Y_{it} = \alpha_i + \gamma_t + f(z_{it}) + \delta x_{it} + \epsilon_{it} \tag{1}$$

where  $Y_{it}$  denotes different outcome variables, e.g. the probability of being employed 12 months after initiating the unemployment spell,  $\alpha_i$  is an individual fixed effect,  $\gamma_t$ is year dummies,  $z_{it}$  is the wealth holdings of the individual,  $x_{it}$  contains the control variables, and  $\epsilon_{it}$  is the error term.

<sup>&</sup>lt;sup>12</sup>HTM individuals are those who are at their overdraft limit at the end of the period-pay and we define that limit to be the individual average monthly income. Because we observe liquid wealth on the first day of a pay-period in the data, HTM are those who have liquidity below their monthly income minus the overdraft limit. Given our definition, this turns up to be those who have negative net liquid wealth the 31st of December.

<sup>&</sup>lt;sup>13</sup>They find that between 25 and 40% of US households are HTM with one-third being poor HTM and two-thirds being wealthy HTM. For Australia, France, Italy, and Spain the number of HTM is below 20% and still the share of wealthy HTM is larger than poor HTM.

Our main coefficients of interest are the effects of the portfolio composition,  $z_{it}$ , on job search behavior. Theory would suggest that individuals with higher wealth levels search less for a job, as they can use some of their savings to smooth consumption. This gives them more time to search for the best job, so besides affecting the unemployment duration directly, it could result in a better job match after unemployment.

A main threat to causal identification of the effects of the portfolio choices is selection; both into job search behaviors and wealth accumulation. As argued above, our estimation strategy eliminates any constant unobserved differences between individuals, but there could be other possible threats to the identification strategy.

In general, we would expect that selection would work in the opposite direction of economic theory. While economic theory would suggest that those with high levels of wealth would find a job at a slower rate, it is probably the case that these individuals have better unobserved labor market characteristics than those with low levels of wealth. Again, using a fixed effects approach takes care of part of this problem by comparing the same individual across different unemployment spells. However, it is still important to have very detailed controls for labor market quality of a given worker.

Our main outcome is the probability of being re-employed 12 months after the beginning of the unemployment spell, but we will also look at 6, 9, 24, and 36 months to check that the choice of 12 months does not drive the results. Appendix A investigates the effects of wealth on the probability of exiting unemployment instead of the probability of becoming re-employed. The results are quantitatively similar to those found using re-employment, but the effects are smaller.

Finally, differences in job search behavior might affect the quality of the resulting job match. To examine this, we proxy the quality of the job following an unemployment spell by looking at the probability that the job or employment spell lasts more than two years. We also look at the re-employment wage in the first job after unemployment.

### 4 Results

In the following, we estimate the effects of individual wealth on the job finding probability step by step. First, we distinguish between liquid net wealth (bank deposits and stock and bond holdings minus debt to credit institutions) and illiquid net wealth (value of properties minus mortgage debt). We proceed by splitting each type of wealth into assets and liabilities. We look at both linear and nonlinear effects of each of the components and include interest payments for each group. We look at the effect of wealth on the probability to become re-employed and on the match quality in terms of wages, job, and employment durations conditional on re-employment.

#### 4.1 Decomposing Wealth

We estimate equation (1) where the outcome is the probability of being re-employed after 12 months. We allow the function of the wealth variables to have a dummy for zero and a linear and quadratic term. In the results presented in this section we only focus on the linear term, and in the next section we will go deeper into the nonlinear effects.

We begin by considering the effect of total net wealth. The first column in Table 4 shows that there is no significant effect of total net wealth on the probability of reemployment. This is a robust finding. We find no effects even if we re-estimate the model for different groups (by gender or marital status for example) or if we look at the probability to find a job after two or three years.

Notice that this is a very precisely estimated zero effect. A one standard deviation increase in total net wealth decreases the probability of finding a job by only 0.06 percent, and we reject effects lower than 0.46 percent at a 95 percent level. At first sight this result is surprising since it contradicts standard economic theory.

However, there are several possible explanations. First, it might be that even our multitude of control variables and using individual fixed effects are not enough to control for differences between individuals with different values of net wealth. In this case, we would expect that individuals with a high level of net wealth also have better unobserved labor market characteristics, which would bias the estimate in a positive direction. However, given our set of control variables, we think that this is only a small part of the explanation. Second, economic theory suggests that workers who are better able to smooth consumption find a job at a slower rate, since they put less effort into finding a job and because they accept fewer of the jobs they find. However, total net wealth might not be a good measure of the ability to smooth consumption. For example, if much of the wealth is illiquid, then accessing it will be costly. Thus, the effect of wealth on the probability of finding a job probably varies depending on the type of portfolio composition in terms of liquidity that the individual has access to.<sup>14</sup> As such, total net wealth is a poor measure for the ability to smooth consumption. Pooling liquid and illiquid net wealth thus renders the identification of any wealth effect very fragile.

In the second column in Table 4 we decompose total net wealth into liquid and illiquid net wealth. Interestingly, we find no effect of illiquid net wealth, but we do find a significantly negative effect of liquid net wealth on the probability of being re-employed one year after the entry into unemployment. Notice however that the estimated effect is still quite small with a one standard deviation increase in liquid net wealth decreasing the probability of re-employment by only 0.2 percent. This result is consistent with standard job search models where the cost of unemployment is higher for workers with lower levels of savings, since they cannot maintain consumption to the same degree as workers who

 $<sup>^{14}</sup>$ For example, one strand of the literature has developed around the heterogeneity in the response to fiscal stimulus, see Kaplan, Violante, and Weidner (2014), Kaplan and Violante (2014), or Misra and Surico (2014)

	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Total wealth	-0.001 (0.002)							
Liquid wealth		$-0.006^{**}$ (0.002)						
Illiquid wealth		(0.002)						
Liquid assets		~	-0.022***	-0.026***	-0.017***	$-0.016^{**}$	-0.023***	-0.022***
Illionid assets			(0.004)-0.003	(0.003)	(0.002) -0.010***	(0.002)	(0.002)- $0.007***$	(0.004)-0.003
··· · · ···· · ···· <b>T</b>			(0.002)	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)
Liquid liabilities			-0.004	-0.000	-0.002	$0.006^{**}$	0.002	-0.004
			(0.004)	(0.003)	(0.003)	(0.002)	(0.002)	(0.004)
Illiquid liabilities			0.001	$-0.011^{***}$	$-0.019^{***}$	$-0.014^{***}$	$-0.015^{***}$	0.001
			(0.004)	(0.003)	(0.003)	(0.002)	(0.002)	(0.004)
Observations	1,398,727	1,398,727	1,398,727	1,398,727	1,398,727	1,398,727	1,398,727	1,398,727
Personal controls	×	x	x		х	х	x	Х
Labor market controls	x	x	x			x	x	х
Financial controls	X	х	х				х	Х
Individual FE	×	X	Х					X

TABLE 4: THE EFFECT OF WEALTH ON THE BROBABILITY TO FIND A LOB WITHIN 13 MONTHS

within the first twelve months as the outcome variable. Wealth variables are expressed in million DKK. Squared terms and zero dummies for the wealth variables as well as month and year dummies are added to all regressions. Personal controls: Age, age<sup>2</sup>, female, single vs. married/cohabiting, educational level, benefit level proxy. Labor market controls: Experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years. Financial controls: Total income, wage income each year for the last three years, hourly wage each year for the last three years. Note: \*\*

can use their savings as a buffer. This is the case in Lentz (2009), who finds a positive relationship between liquid net wealth and unemployment durations. In his model, this is a result of the search behavior of the workers as he finds that wealthier workers search less. Consistent with the results presented here, Lentz (2009) finds no significant effects of being a home owner.

Liquid and illiquid net wealth pool assets and liabilities in each category. Economically, the two can have very different effects. For example, two individuals with a total net wealth of 10,000 DKK where one has liquid assets worth 10,000 DKK and the other has 200,000 DKK in liquid assets and 190,000 DKK in liquid liabilities (debt to e.g. banks) are likely to have different accesses to credit. In that case, their ability to smooth consumption differs, and liquid net wealth can be misleading. Moreover, the distinction between assets and liabilities is necessary if the interest rates on liquid liabilities and on cash holdings or other liquid assets are different. Usually the interest rate on bank deposits is lower than the interest rate paid to credit institutions or on bank loans.

The third column in Table 4 provides results when assets and liabilities are separated for both the liquid and illiquid categories. As in column (1) and (2), we use our full set of controls. Again, in line with the job search theory, liquid assets have a significant and negative effect on the probability of being re-employed one year after the beginning of an unemployment spell. Job seekers with more liquid assets at the entry into unemployment have a lower probability of becoming re-employed, and the effect is almost five times what we found when using liquid net wealth. Because liquid net wealth pools liquid liabilities and liquid assets, an increase in liquid net wealth can come from a change in both sides of the worker's financial position, either an increase in the asset holdings or a decrease in liabilities. On the contrary, separating the two allows us to isolate the effect of an increase in liquid assets. Here, an increase in the amount of liquid assets by 1 million DKK/150.000 USD (one standard deviation) decreases the probability of becoming re-employed within the first year by 2.2 (0.5) percentage points. The result that higher liquid assets increase the time spent in unemployment is also found in the papers by Card, Chetty, and Weber (2007) and Basten, Fagereng, and Telle (2014). Both papers exploit a discontinuity in the eligibility to severance payments in Austria and Norway, respectively to create liquidity shocks at the entry into unemployment. In both papers, the increase in liquid assets increases the duration of non-employment. Table 4 shows that there are no significant effects of debt to banks and credit institutions (liquid liabilities), the value of properties (illiquid asset), or of mortgage debt (illiquid liabilities).

Columns (4) to (8) show the estimated coefficients where different sets of control variables as well as the individual fixed effect are added. Column (8) is the same as column (3). We show this to highlight the importance of using the fixed effect approach even after controlling for a vast set of observable differences. Looking at the effect of liquid assets, we see that once we control for personal characteristics, the effect is rather stable across specifications. For illiquid assets and illiquid liabilities, we see that the

effect of higher property value and higher mortgage debt both decrease when adding more control variables and vanish once the fixed effect is included in the regression. Not taking the unobserved, time-invariant individual heterogeneity into account would lead to different conclusions about the effects of these two variables. The effect of liquid liabilities disappears once all control variables are included. These results show that the inclusion of a vast set of control variables is important and furthermore, it is important to account for unobserved individual heterogeneity.

#### 4.2 Nonlinear Effects Of Wealth

In the previous section, we investigated the linear effects of the different components of wealth on the probability to find a job. However, economic theory would suggest nonlinear effects to be important, as well as for example poor individuals would respond more than wealthy individuals to changes in the wealth components. To include this in the estimation model, we specify a linear spline for each of the wealth components. The knots are chosen in the following way. For each wealth component we create a knot at the 99th percentile, where the percentiles are taken over the non-zero values of that component. We then create intervals  $(0, \overline{x}_{99}/5], (\overline{x}_{99}/5, 2\overline{x}_{99}/5], ..., (4\overline{x}_{99}/5, \overline{x}_{99}])$ , where  $\overline{x}_{99}$  is the value of the 99th percentile. Thus, we have 7 groups: One with zero, five in each of the intervals, and one group with values higher than the 99th percentile. We then create a spline with knots between the intervals and at the 99th percentile and add a dummy for zero. I.e. for liquid assets,  $la_{it}$ , we let

$$f(la_{it}) = \beta_0 \mathbb{1}[la_{it} = 0] + \beta_1 la_{it} + \beta_2 (la_{it} - \overline{la}_{99}/5) \mathbb{1}[la_{it} > \overline{la}_{99}/5] + \dots + \beta_6 (la_{it} - \overline{la}_{99}) \mathbb{1}[la_{it} > \overline{la}_{99}]$$
(2)

where  $\overline{la}_{99}$  is the value of the 99th percentile of liquid assets. Thus, the function is continuous except at the value zero, and the unit of measurement is again DKK million. We do the same for all four wealth components and estimate the same model as table 4, column (8), but with the nonlinear specification for the wealth components. Figure 2 plots the marginal effects for each group and 95 percent confidence bands. Notice, that for group 0 it is the effect of having zero of that particular wealth component.<sup>15</sup>

The first graph shows the results for liquid assets, and we see stronger marginal effects (in absolute value) at the bottom of the wealth distribution: for individuals with low levels of liquid assets, the effect of additional liquidity is higher in the sense that they tend to find a job with a lower probability. This is in accordance with standard economic theory where a small amount of liquid assets for people who are close to their borrowing constraint decreases the effort to get out of unemployment. It also decreases the acceptance probability as the additional assets increase the possibility for consumption

 $<sup>^{15}{\</sup>rm Figure~3}$  in the Appendix shows the results when using dummy groups instead of the spline. The results show a similar pattern.

FIGURE 2: MARGINAL EFFECTS OF ASSETS AND LIABILITIES ON THE PROBABILITY TO BE RE-EMPLOYED BEFORE 12 MONTHS.



Note: This figure displays the estimated marginal effects and 95 percent confidence bands where the four wealth groups are split into groups. The outcome variable is finding a job in the first twelve months after the beginning of unemployment. Month and year dummies as well as individual fixed effects are added. All controls are added (age, age<sup>2</sup>, female, single vs. married/cohabiting, educational level, benefit level proxy, experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years, total income, wage income each year for the last three years, and hourly wage each year for the last three years).

smoothing.

For the other three groups we find mixed evidence. There is a tendency that at low levels of liquid liabilities there is a marginally significant negative effect on the reemployment probability, but the effect is qualitatively small. One reason why we do not find that liquid liabilities matter for the re-employment probability could be that workers with better access to credit in general are granted bigger loans. Thus, the variable picks up two competing forces. First, the fact that the individual has a big loan in the bank means that the individual is more credit constrained and therefore should search more for a job. Second, since the individual was in the first place able to borrow in the bank, the individual might have better access to credit again because of unobserved heterogeneity. For both illiquid liabilities and illiquid assets we find no clear marginal effects.

#### 4.3 Interest Payments

Besides the wealth components themselves, the possible interest rates earned or paid could also play a part. The size of a loan in itself might not be a good indicator of the constraints on consumption. What really matters is the financial conditions of the loan, i.e. its duration and the required monthly payments, not the value of debt in itself. Following that idea, we introduce in our estimations three new variables; the interest payments on mortgage, the interest payments to financial institutions (that is on liquid liabilities), and the interest payments the worker receives from her liquid assets. As usual, the variables are defined the year before unemployment. Table 5 displays the results, where wealth enters linearly, interest payments are included in column (1)-(3), and the results where wealth enters nonlinearly and interest payments are included in column (4)-(6).<sup>16</sup> The first column of each specification shows the results without control variables, the second column adds control variables, and the third column further adds the individual fixed effect to see how that affects the results.

TABLE 5: THE EFFECTS OF INTEREST PAYMENTS ON THE PROBABILITY TO BECOME RE-EMPLOYED WITHIN THE FIRST 12 MONTHS AFTER UNEMPLOYMENT BEGINS.

	(1)	( <b>2</b> )	( <b>3</b> )	(4)	(5)	(6)
	(1)	(2)	(0)	(4)	(0)	(0)
	1	linear wealt.	h	No	n-linear wea	lth
Positive interest payments,	-0.028	0.131	0.168	0.018	0.154	0.202
banks (liquid assets)	(0.119)	(0.107)	(0.136)	(0.110)	(0.107)	(0.132)
Negative interest payments,	$0.654^{***}$	0.080	$0.244^{***}$	$0.570^{***}$	0.013	$0.363^{***}$
banks (liquid liabilities)	(0.080)	(0.050)	(0.076)	(0.095)	(0.053)	(0.079)
Negative interest payments,	-0.262***	-0.379***	-0.043	-0.240***	$-0.341^{***}$	$-0.122^{**}$
mortgage institutions (illiquid liabilities)	(0.042)	(0.040)	(0.058)	(0.044)	(0.042)	(0.059)
Liquid assets	-0.024***	-0.025***	-0.024***			
	(0.003)	(0.003)	(0.005)			
Illiquid assets	-0.017***	-0.008***	-0.003			
	(0.001)	(0.001)	(0.002)			
Liquid liabilities	-0.019***	0.001	-0.011**			
	(0.003)	(0.003)	(0.005)			
Illiquid liabilities	-0.003	-0.002	0.003			
	(0.003)	(0.003)	(0.005)			
Observations	$1,\!398,\!727$	$1,\!398,\!727$	$1,\!398,\!727$	$1,\!398,\!727$	$1,\!398,\!727$	$1,\!398,\!727$
All controls		х	х		х	х
Individual FE			х			х

Note: \*\*\*, \*\*, and \* indicate statistical significance at the 1, 5, and 10 percent levels, respectively. This table displays the results from estimations of Equation 1 using the probability of becoming reemployed in the first twelve months after the unemployment spell begins as the outcome variable. Wealth variables are as specified in equation (2). Interest payments are expressed in millions of Danish kroner. Month and year dummies are added to all regressions. Personal controls: Age, age<sup>2</sup>, female, single vs. married/cohabiting, and educational level. Labor market controls: Experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years. Financial controls: Total income, wage income each year for the last three years, hourly wage each year for the last three years.

When the wealth components enter linearly, column (1) shows that higher negative

<sup>&</sup>lt;sup>16</sup>For the nonlinear specification, only the effects of interest payments are shown in the table. The nonlinear effects of the four wealth components when interest payments are included can be seen in Figure 4 and 5 in the Appendix. The results do not change much compared to Figure 2 above and Figure 3 in the Appendix.

interest payments to banks increase the probability of re-employment, whereas mortgage interest payments decrease the probability. Adding our controls and the individual fixed effect removes the effect of mortgage payments, but the effect of interest payments to banks remains (column (3)). Comparing the effects of the wealth components here to those found in Table 4 (columns (4), (7) and (8)) shows that adding the interest payments does not affect the estimated coefficients of the wealth components much.

Turning to the nonlinear specification in Equation 2 (columns 4 to 6), we see that interest payments on liquid debt which reduces the resources available for consumption speed up the exit from unemployment. The estimated effect is sensitive to the inclusion of control variables and unobserved individual heterogeneity, but the effect remains significant. This is in accordance with economic theory, and it is worth noticing that this effect is an order of magnitude larger in absolute value than the effect of liquid asset. An increase of yearly interest payments by 1 million DKK (150.000 USD) would increase the probability of finding a job in the first year by 36 percentage points.

Interest payments on mortgage debt are found to have opposite effect (but only at a 5 percent level). The difference between the effect of interest rates on liquid and illiquid liabilities is interesting in itself. It could be caused by the fact that owning a house, which is almost always financed by a mortgage, reduces job finding rates, since geographical mobility is costly. This is consistent with empirical evidence on the effect of home ownership on job search (e.g. Munch, Rosholm, and Svarer, 2008, Head and Lloyd-Ellis, 2012, Caliendo, Gielen, and Mahlstedt, 2015, or le Maire and He, 2018).

#### 4.4 Wealth Effects on Match Quality

We now turn to look at what happens after re-employment. The worker's financial wealth at the entry into unemployment can impact the next employment spell because it impacts the type of job a worker is willing to accept.<sup>17</sup> Empirically, the evidence is scarce. Using a quasi-experiment and Austrian data, Card, Chetty, and Weber (2007) find no evidence of match quality gains for workers who receive a positive liquidity shock at the entry into unemployment. The same is the case in Basten, Fagereng, and Telle (2016), using a lump-sum severance payment in Norway and finding no effects on the length of the new job or the re-employment wage. Confirming the elusive nature of the link between unemployment value and reservation wage/utility, the results on the effect of unemployment insurance generosity on reservation wages and job stability are also mixed (see references in Tatsiramos (2014)). One interesting exception is Luo and Mongey (2019) who find a positive effect of student debt on wage after graduation - a very particular type of debt at a very particular time in the career.

Table 6 presents the results for three measures of a job's match quality. The prob-

 $<sup>^{17}</sup>$ Notice that, even in theory, this is not always true. If the marginal job arrival rates are the same in unemployment and employment, then the reservation wage simply equals the unemployment benefits (Lise (2013)).

	(1)	(2)	(3)
	First job	Re-employment spell	Re-employment
	above 1 year	above 2 years	wage
Liquid assets	-0.008	-0.007	-0.002**
	(0.007)	(0.007)	(0.001)
Illiquid assets	0.001	0.006	-0.001***
	(0.003)	(0.004)	(0.000)
Liquid liabilities	-0.006	-0.025***	-0.000
	(0.009)	(0.009)	(0.001)
Illiquid liabilities	0.019**	0.004	$0.001^{*}$
	(0.008)	(0.008)	(0.001)
Observations	846,244	$798,\!596$	881,900
All controls	х	х	х
Individual FE	х	х	х

TABLE 6: THE EFFECTS OF ASSETS AND LIABILITIES ON THE MATCH QUALITY OF THE JOB FOLLOWING UNEMPLOYMENT.

Note: \*\*\*, \*\*, and \* indicate statistical significance at the 1, 5, and 10 percent levels, respectively. Results from estimations of Equation 1 using the probability of the first job accepted lasting 12 months or more, the probability of the re-employment spell lasting 24 months or more and the re-employment labor income as the outcome variables and controlling for individual fixed effects. Wealth variables are expressed in million DKK. Squared terms and zero dummies for the wealth variables, individual fixed effects as well as month and year dummies are added to all regressions. All controls are added (age, age<sup>2</sup>, female, single vs. married/cohabiting, educational level, experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years, total income, wage income each year for the last three years, and hourly wage each year for the last three years).

ability that the first job accepted lasts more than one year, the probability that the re-employment spell (thus including job-to-job mobility) lasts more than two years, and finally the re-employment wage. Effects on the duration of the first job and on the duration of the entire re-employment spell are conceptually different. Indeed, if job-to-job mobility is risky, more liquid individuals might be more willing to change jobs and thus could have shorter durations in the first job after unemployment, but longer employment duration on average because they reallocate.

We find very mixed results. None of the variables of interest have an effect on all our three measures of a job's match quality, and only some have effects at a one percent significance level. However, if we concentrate on the most significant results, the picture is quite consistent with what we have already found. First, the value of the properties (illiquid asset) reduces the re-employment wage. Again, the effect is small but in accordance with the idea that being a home owner could actually reduce mobility on the labor market by increasing the adjustment costs implied by a job change (see Munch, Rosholm, and Svarer (2008), Winkler (2011), Head and Lloyd-Ellis (2012), Caliendo, Gielen, and Mahlstedt (2015)). Second, we find evidence that liquid liabilities could render re-employment spell more fragile since they decrease the probability for the spell to be above 2 years. Third, there is no effect of liquid assets on the probability of the first job lasting at least one year or the employment spell at least lasting two years. This may seem surprising at first, but this lines up with the results found in Card, Chetty, and Weber (2007) and Basten, Fagereng, and Telle (2014): liquidity impacts exit rates, but not the match quality.

#### 4.5 Robustness

In this section we show that our main results are robust to a number of choices that we have made.

**Timing of Job Finding** First, we investigate how important the choice of the outcome variable regarding the number of months before becoming re-employed is. As Table 3 showed, 69% of the sample are re-employed after 6 months, 78% after 9 months, while 82%, 90%, and 93% are re-employed after one, two, and three years. Table 7 shows the effect of the four wealth components on the probability of becoming re-employed after 6, 9, 12, 24, and 36 months. Notice that 12 months is the baseline as shown in Table 4 and that we only show the preferred specification with a full set of controls corresponding to column (3) in Table 4.

	(1)	(2)	(3)	(4)	(5)
		Re-emp	oloyed within	(months)	
	6 months	9 months	12  months	24  months	36  months
Liquid assets	-0.029***	-0.028***	-0.022***	-0.012***	-0.008**
	(0.005)	(0.004)	(0.004)	(0.004)	(0.004)
Illiquid assets	-0.007***	-0.007***	-0.003	0.003	0.001
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Liquid liabilities	-0.009**	-0.006	-0.004	0.005	0.005
	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
Illiquid liabilities	-0.002	0.004	0.001	-0.008**	-0.005*
	(0.005)	(0.004)	(0.004)	(0.003)	(0.003)
Observations	1,440,899	$1,\!426,\!558$	1,398,716	1,293,248	1,161,330
All controls	х	х	х	х	х
Individual FE	х	х	Х	х	Х

TABLE 7: THE EFFECTS OF ASSETS AND LIABILITIES ON THE PROBABILITY TO HAVE FOUND RE-EMPLOYMENT AFTER 6, 9, 12, 24, OR 36 MONTHS.

Note: \*\*\*, \*\*, and \* indicate statistical significance at the 1, 5, and 10 percent levels, respectively. Results from estimations of Equation 1 using re-employment in the first 6, 9, 12, 24, or 36 months as the outcome variable. Wealth variables are expressed in million DKK. Squared terms and zero dummies for the wealth variables, individual fixed effects as well as month and year dummies are added to all regressions. All controls are added (age, age<sup>2</sup>, female, single vs. married/cohabiting, educational level, experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years, total income, wage income each year for the last three years, and hourly wage each year for the last three years).

Generally, the choice of how many months after the beginning of the unemployment spell does not affect the results qualitatively. The negative effect of higher liquid assets is strongest for the probability of being re-employed after 6 months and then slowly drops from 2.8 percentage points to 1 percentage point after three years. As expected, these results indicate that the effect of initial wealth is decreasing as the duration of unemployment increases. When a worker has been unemployed for a longer period of time, the effect of wealth at the beginning of the unemployment spell diminishes. Higher illiquid assets decrease the re-employment probability the first month but doesn't seem to impact the re-employment probability after one year. There doesn't seem to be any convincing effects when considering liabilities.

**Heterogeneity Across Subgroups** We would expect that the results of wealth vary across subgroups. We have thus chosen to split on marital status, education, wage in last job, and home ownership. The results are shown in Table 8 where we re-do the estimations where the four wealth components enter linearly. Column (1) depicts the main results from Table 4 column (3).

The results in Section 4.1 do not control for the fact that individuals can pool risks within households. In practice, risk sharing within households is very common, and it is expected to have important consequences for consumption, income and labor supply, or the impact of social security (e.g. Ortigueira and Siassi (2013), Blundell, Graber, and Mogstad (2015), or Yum (2018) for recent references). However, there are few direct measures of how labor supply decisions are affected by spouse's wealth. Most of the papers generally pool wealth at the household level while, in principle, sharing is never perfect and can be subject to intra-household bargaining. Column (2) shows the results separately for singles, while (3) shows the results for married or cohabiting couples.

There is almost no difference between being single or married/cohabiting for the effect of liquid assets. For both groups the effect is almost the same as for the entire sample. The same is seen for the other wealth components where we do not see any significant differences. Turning to spouses' wealth, we find that if the spouse has a lot of liquid assets, then this decreases the probability of finding a job for the individual. A 1 million DKK (150.000 USD) increase in a spouse's liquid assets decreases the probability of reemployment for the worker by 0.7 percentage points, and a similar increase in the spouse's illiquid assets decreases the probability by 0.5 percentage points. Thus, the effects are lower than the direct effect of the worker's own liquid assets, indicating that wealth sharing among spouses is not perfect.

Next, we split the sample into three educational groups. The table shows that no matter the degree of education, liquid assets have a negative effect on the probability of becoming re-employed within the first year after the unemployment spell begins. Interestingly, the effect is highest for the low and highly educated groups. For either of these groups, a 1 million DKK (150.000 USD) increase in liquid assets decreases the probabil-

BLE 8:	THE EFFECTS OF ASSETS AND LIABILITIES ON THE PROBABILITY TO BE RE-EMPLOYED BEFORE 12 MONTHS FOR SUB-	
UPS.		

	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
		Marital s	tatus		Education		M	age	Home	owner
	All	Single	Married/cohab.	$\operatorname{Low}$	Medium	High	$\operatorname{Low}$	High	Yes	$N_{O}$
Liquid assets	-0.022***	-0.023***	-0.021***	-0.030***	-0.020***	-0.027***	-0.030**	$-0.021^{***}$	-0.008*	-0.060***
	(0.004)	(0.008)	(0.005)	(0.010)	(0.006)	(0.008)	(0.013)	(0.005)	(0.005)	(0.014)
Illiquid assets	-0.003	0.004	-0.004	$-0.012^{**}$	$0.007^{**}$	-0.004	-0.006	-0.004	-0.004	
	(0.002)	(0.004)	(0.003)	(0.005)	(0.003)	(0.005)	(0.007)	(0.003)	(0.003)	
Liquid liabilities	-0.004	-0.010	-0.007	-0.000	-0.008	-0.009	-0.015	0.000	$-0.012^{**}$	0.004
	(0.004)	(0.009)	(0.005)	(0.009)	(0.006)	(0.010)	(0.012)	(0.005)	(0.006)	0.008
Illiquid liabilities	0.001	0.004	-0.003	$0.019^{**}$	$-0.013^{**}$	0.009	$0.029^{**}$	-0.005	-0.009**	0.046
	(0.004)	(0.009)	(0.005)	(0.010)	(0.006)	(0.000)	(0.012)	(0.005)	(0.005)	(0.031)
Spouse, liquid assets			-0.007**							
			(0.003)							
Spouse, illiquid assets			-0.005***							
			(0.002)							
Spouse, liquid liabilities			-0.000							
			(0.003)							
Spouse, liquid illiabilities			-0.004							
			(0.003)							
Observations	1,398,727	581,919	816,797	521,119	662, 342	215, 255	600, 157	798,559	599,813	798,903
All controls	х	X	Х	х	х	х	х	х	х	x
Individual FE	Х	х	х	х	Х	Х	Х	х	х	х
Note: ***, **, and * indicate st	tatistical signi	ficance at the	e 1, 5, and 10 percent	levels, respec	tively. Result	s from estima	ations of Equ	lation 1 using	re-employm	ent
in the first twelve months as th	le outcome val	riable. Wealtl	h variables and intere	st payments	are expressed	in million D	KK. Squared	terms and ze	ero dummies	for
the wealth variables, individual	l fixed effects a	us well as mon	nth and year dummie	s are added t	o all regressio	as. All contre	ols are added	(age, age <sup>2</sup> , fe	emale, single	vs.
married/cohabiting, educationa	al level, experi	ence, tenure i	in previous job, occur	oation, memb	er of unemplo	yment insura	nce fund, ex	perience gaine	ed each year	for
the last three years, degree of 1	unemployment	c each year fo	or the last three year:	s, total incom	ie, wage incor	ne each year	for the last <sup>1</sup>	chree years, a	und hourly w	age
each year for the last three year	$\operatorname{rs}).$									

ity of re-employment by around 2-3 percentage points. On the other hand, for those with low or medium educations, both illiquid assets and illiquid liabilities now affect the probability significantly (but only at a 5% level). An increase in the property value increases the re-employment probability, whereas an increase in mortgage debt decreases the probability.

Splitting by pre-unemployment wages in columns (7) and (8) also affects the results. Those with hourly wages below the median before unemployment have a higher decrease in the re-employment probability compared to the rest. Furthermore, for the individuals with low wage, we see a positive effect of mortgage debt.

Finally, in columns (9) and (10) we split the sample by home ownership. Interestingly, the effect of liquid assets is primarily driven by non-homeowners.

**Changes in Sample** In Table 9, we show how the main results differ across different samples which are created to mitigate various concerns. Column (1) shows the full sample results.

The first concern is that individuals in our sample might have different potential lengths of unemployment insurance.

As an example, consider an individual who has an unemployment spell of a given length and then finds a job. The individual looses the job after a short period of time and has not become re-eligible for full unemployment insurance during the new unemployment spell. This implies that the effective potential length of the new unemployment insurance period is shorter than the maximum duration due to the initial draw on the length of the first unemployment spell. To alleviate this concern, we condition on short prior durations of unemployment in columns (2) and (3). This does not seem to matter much for the result.

Another concern is that there is a difference between when we measure the wealth variables and the start of the unemployment spell. This is due to the fact that wealth is only measured at an annual frequency in the registers. In particular, it is measured on 31 December each year. In column (4) we select only unemployment spells that started during the first six months of the year, such that wealth was measured closer to the beginning of the spell. This reduces the estimate, but it is still significant at a 1 percent level. Liquid liabilities also become significant at a 5 percent level.

Finally, we split the sample based on the business cycles. In column (5) we exclude observations from the Financial Crisis. This decreases the effect of liquid assets on job finding to -0.032. This indicates that during the Financial Crisis, liquid assets did not matter as much for job finding which could be caused by the overall deterioration of the job finding prospects.

	(1)	(2)	(3)	(4)	(5)
		U during p	ast 3 years	U start	Business cycles
	Full sample	< 6 months	< 18 months	first 6 months	Excl. 2008Q2-2010Q3
iquid assets	-0.022***	-0.015***	-0.018***	-0.014***	-0.032***
	(0.004)	(0.005)	(0.004)	(0.008)	(0.005)
lliquid assets	-0.003	-0.003	-0.002	-0.001	-0.008***
	(0.002)	(0.003)	(0.002)	(0.004)	(0.003)
iquid liabilities	-0.004	-0.008	-0.003	$-0.019^{**}$	$-0.013^{**}$
	(0.004)	(0.005)	(0.004)	(0.008)	(0.005)
lliquid liabilities	0.001	0.001	0.001	-0.000	-0.002
	(0.004)	(0.005)	(0.004)	(0.008)	(0.004)
)bservations	1,398,727	955,444	1,326,251	419, 342	1,007,196
All controls	х	х	х	Х	Х
ndividual FE	Х	Х	Х	Х	х

TABLE 9: ROBUSTNESS CHECKS OF MAIN RESULTS.

within the first twelve months as the outcome variable. Wealth variables are expressed in million DKK. Squared terms and zero dummies for the wealth variables Note: \*\*\*, \*\*, and \* indicate statistical significance at the 1, 5, and 10 percent levels, respectively. Results from estimations of Equation 1 using re-employment as well as month and year dummies are added to all regressions. Personal controls: Age, age<sup>2</sup>, female, single vs. married/cohabiting, and educational level. Labor market controls: Experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years. Financial controls: Total income, wage income each year for the last three years, hourly wage each year for the last three years.

## 5 Conclusion

This is one of the first papers to show that the portfolio *composition* of the unemployed workers affects their job search behavior. The decomposition of wealth into liquid and illiquid assets and liabilities is important to understand how wealth affects the exit rate from unemployment. In particular, we have shown that higher liquid assets decrease the probability of becoming re-employed within the first year after the beginning of the unemployment spell, and that the effect is quantitatively very different from the one found using net wealth. The effects are nonlinear in the sense that the marginal effect of liquid assets is more negative for low levels of assets as expected. Moreover, we find evidence that interest payments to banks, when coming from credit card debt or consumption credits, speed up re-employment.

The results are stable across educational levels, the pre-unemployment wage level as well as the marital status of the individuals. However, the results on liquid assets seem to be driven by non-homeowners.

As liquid assets prolonged the time spent in unemployment, we investigated whether this had any effect on the subsequent job match quality. This could be expected, as individuals with higher liquid assets could spend more time searching for the best job match while still maintaining a high consumption level. Similar to previous research on the subsequent job quality, we do not find any clear pattern of liquid asset on the match quality of the job after re-employment but some (quantitatively limited) negative effects for liquid liabilities.

Overall, these results offer a nuanced picture of the effect of wealth on job search. Both the decomposition by degrees of liquidity and the separation between asset, debt, and interest payments are important.

## References

- ALEXOPOULOS, M., AND T. GLADDEN (2004): "The Effects of Wealth, and Unemployment Benefits on Search Behavior and Labor Market Transitions," Econometric Society 2004 North American Summer Meetings 517, Econometric Society.
- ALGAN, Y., A. CHERON, J.-O. HAIRAULT, AND F. LANGOT (2003): "Wealth Effect on Labor Market Transitions," *Review of Economic Dynamics*, 6(1), 156–178.
- BASTEN, C., A. FAGERENG, AND K. TELLE (2014): "Cash-on-hand and the Duration of Job Search: Quasi-experimental Evidence from Norway," *Economic Journal*, 0(576), 540–568.
- (2016): "Saving and Portfolio Allocation Before and After Job Loss," *Journal* of Money, Credit and Banking, 48(2-3), 293–324.
- BLOEMEN, H., AND E. STANCANELLI (2001): "Individual Wealth, Reservation Wages, and Transitions into Employment," *Journal of Labor Economics*, 19(2), 400–439.
- BLUNDELL, R., M. GRABER, AND M. MOGSTAD (2015): "Labor income dynamics and the insurance from taxes, transfers, and the family," *Journal of Public Economics*, 127(C), 58–73.
- CALIENDO, M., A. C. GIELEN, AND R. MAHLSTEDT (2015): "Home-ownership, unemployed's job search behavior and post-unemployment outcomes," *Economics Letters*, 137(C), 218–221.
- CARD, D., R. CHETTY, AND A. WEBER (2007): "Cash-On-Hand and Competing Models of Intertemporal Behavior: New Evidence from the Labor Market," *Quarterly Journal of Economics*, 122(4), 1511–1560.
- COULSON, N. E., AND L. M. FISHER (2009): "Housing tenure and labor market impacts: The search goes on," *Journal of Urban Economics*, 65(3), 252–264.
- GROSS, D. B., AND N. S. SOULELES (2002): "Do Liquidity Constraints and Interest Rates Matter for Consumer Behavior? Evidence from Credit Card Data," *The Quarterly Journal of Economics*, 117(1), 149–185.
- HEAD, A., AND H. LLOYD-ELLIS (2012): "Housing Liquidity, Mobility, and the Labour Market," *Review of Economic Studies*, 79(4), 1559–1589.
- KAPLAN, G., AND G. L. VIOLANTE (2014): "A Model of the Consumption Response to Fiscal Stimulus Payments," *Econometrica*, 82(4), 1199–1239.
- KAPLAN, G., G. L. VIOLANTE, AND J. WEIDNER (2014): "The Wealthy Hand-to-Mouth," *Brookings Papers on Economic Activity*, 45(1 (Spring)), 77–153.

- LE MAIRE, D., AND A. X. HE (2018): "How Does Liquidity Constraint Affect Employment and Wages? Evidence from Danish Mortgage Reform," mimeo, University of Copenhagen.
- LENTZ, R. (2009): "Optimal Unemployment Insurance in an Estimated Job Search Model with Savings," *Review of Economic Dynamics*, 12(1), 37–57.
- LENTZ, R., AND T. TRANAS (2005): "Job Search and Savings: Wealth Effects and Duration Dependence," *Journal of Labor Economics*, 23(3), 467–490.
- LISE, J. (2013): "On-the-Job Search and Precautionary Savings," Review of Economic Studies, 80(3), 1086–1113.
- LUO, M., AND S. MONGEY (2019): "Assets and Job Choice: Student Debt, Wages, and Job Satisfaction," 2019 Meeting Papers 1220, Society for Economic Dynamics.
- MISRA, K., AND P. SURICO (2014): "Consumption, Income Changes, and Heterogeneity: Evidence from Two Fiscal Stimulus Programs," American Economic Journal: Macroeconomics, 6(4), 84–106.
- MUNCH, J. R., M. ROSHOLM, AND M. SVARER (2008): "Home ownership, job duration, and wages," *Journal of Urban Economics*, 63(1), 130–145.
- ORTIGUEIRA, S., AND N. SIASSI (2013): "How important is intra-household risk sharing for savings and labor supply?," *Journal of Monetary Economics*, 60(6), 650–666.
- RENDON, S. (2006): "Job Search And Asset Accumulation Under Borrowing Constraints," *International Economic Review*, 47(1), 233–263.
- SHI, S., AND G. CHAUMONT (2017): "Wealth Accumulation, On the Job Search and Inequality," 2017 Meeting Papers 128, Society for Economic Dynamics.
- TATSIRAMOS, K. (2014): "Unemployment benefits and job match quality," *IZA World* of Labor, pp. 1–44.
- TELYUKOVA, I. A. (2013): "Household Need for Liquidity and the Credit Card Debt Puzzle," *Review of Economic Studies*, 80(3), 1148–1177.
- WINKLER, H. (2011): "The Effect of Homeownership on Geographic Mobility and Labor Market Outcomes," 2011 Meeting Papers 196, Society for Economic Dynamics.
- YUM, M. (2018): "On the distribution of wealth and employment," *Review of Economic Dynamics*, 30, 86–105.

# Appendices

## A Probability of Exiting Unemployment

In this appendix we will show how the results change when we use the exit from unemployment as the outcome variable instead of the re-employment probability. When looking at exiting unemployment, individuals can exit to the following states: Employment, Education, Retirement, Other public benefits, or Non-participation. The frequencies for each of the exit states can be seen in Table 10. The table shows that 62% exit to employment

	Frequency	Percent
Employment	858,194	61.36
Education	$17,\!974$	1.29
Non-participation	$122,\!277$	8.74
Other public benefits	$398,\!174$	27.82
Retirement	$11,\!108$	0.79

TABLE 10: EXIT STATES

Note: The table shows the possible exit states after the unemployment spell ends.

whereas 28% exit to other public benefits. Notice that even if the individual exits to other public benefits, the individual could easily find a job later. In Table 11 we show the probabilities of exiting from unemployment within 6, 9, 12, 24, and 36 months as well as the probabilities for becoming re-employed (i.e. exiting to employment) from Table 3.

TABLE 11:	Probability	OF	EXITING	FROM	UNEMPLOYMENT	WITHIN	6,	9,	12,	24,
and 36 mc	NTHS.									

	Mean	SD
Fraction exiting unemployment after (%)		
6 months	78.27	41.24
9 months	88.35	32.09
12 months	93.10	25.34
24 months	99.10	9.42
36 months	99.82	4.22
Fraction re-employed after $(\%)$		
6 months	69.42	46.08
9 months	77.94	41.47
12 months	82.42	38.06
24 months	90.49	29.34
36 months	93.32	25.15

Note: The table shows the probability of exiting from unemployment within 6, 9, 12, 24 and 36 months after the beginning of the unemployment spell.

As can be seen from the table, the fraction exiting unemployment before a given month is of course larger than the fraction exiting to employment. To have a comparable fraction of observations where an exit is observed, Table 12 shows the effect of wealth on the probability of exiting unemployment within 9 months after the unemployment spell begins. The fraction exiting unemployment after 9 months is 88%, whereas the fraction exiting to employment after 12 months which is the main outcome in Section 4 is 82%.

In Table 12 we replicate Table 4, but we use exiting unemployment as the outcome instead of finding a job.

	(1)	(2)	(3)
Total net wealth	-0.000		
	(0.001)		
Liquid net wealth		-0.000	
		(0.002)	
Illiquid net wealth		-0.001	
		(0.002)	
Liquid assets			-0.012***
			(0.003)
Illiquid assets			-0.001
			(0.002)
Liquid liabilities			-0.013***
			(0.003)
Illiquid liabilities			0.001
			(0.004)
Observations	$1,\!426,\!558$	$1,\!426,\!558$	$1,\!426,\!558$
All controls	Х	Х	х
Individual FE	х	х	х

TABLE 12: THE EFFECT OF WEALTH ON THE PROBABILITY OF EXITING UNEMPLOY-MENT WITHIN 9 MONTHS.

Note: \*\*\*, \*\*, and \* indicate statistical significance at the 1, 5, and 10 percent levels, respectively. Results from estimations of Equation 1 using exit within the first nine months as the outcome variable. Wealth variables are expressed in million DKK. Squared terms and zero dummies for the wealth variables as well as month and year dummies are added to all regressions. Personal controls: Age, age<sup>2</sup>, female, single vs. married/cohabitating, and educational level. Labor market controls: Experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years. Financial controls: Total income, wage income each year for the last three years, hourly wage each year for the last three years.

Table 12 shows no effect of total net wealth. This is similar to the results from Table 4. When splitting total net wealth into liquid and illiquid net wealth, no effect on the probability of exiting unemployment is found for either of these variables. In Table 4 we saw that there was a small but significantly negative effect of liquid net wealth on the probability of becoming re-employed. If we look at the effect of the four wealth components on the probability of exiting unemployment within the first 9 months, we see that there is a negative effect of liquid assets. This effect is smaller than the effect of liquid

assets on the probability of re-employment within 12 months, which was found to be 2.2 percentage points. Furthermore, Table 12 shows a negative effect of liquid liabilities on the probability of exiting unemployment. This is only significant at a 5% significance level though. In general, the effects of the wealth components on the probability of exiting unemployment are smaller than the similar effects of the wealth components on the probability of becoming re-employed, but qualitatively the results are similar.

## **B** Nonlinear Effects using Dummy Groups

This appendix shows the nonlinear effects estimating the same regression as for Figure 2, but using dummy groups instead of the spline. We show the results from this specification in Figure 3. Notice, that group 0, which is the reference group, and group 6 are special. Group 0 has zero of the wealth component, while the individuals in group 6 are those with an extreme amount, i.e. higher than the 99th percentile.

FIGURE 3: NONLINEAR EFFECTS OF ASSETS AND LIABILITIES ON THE PROBABILITY TO BE RE-EMPLOYED BEFORE 12 MONTHS USING DUMMIES.



Note: This figure displays the estimated coefficients and 95 percent confidence bands, where the four wealth groups are split into groups, and dummies for each group are used. The outcome variable is the probability of becoming re-employed in the first twelve months after the unemployment spell begins. Month and year dummies as well as individual fixed effects are added. All controls are added (age, age<sup>2</sup>, female, single vs. married/cohabitating, educational level, experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years, total income, wage income each year for the last three years, and hourly wage each year for the last three years).

Looking first at liquid assets, we again find a clear effect. The more liquid assets, the lower the re-employment probability, and the effect is clearly decreasing in the groups. For liquid liabilities we find a significant cumulative effect, i.e. having liquid liabilities in group 5 significantly decreases the re-employment probability compared to being in group 0. But the pattern is less clear than for liquid assets.

Again, for the illiquid variables we find no clear pattern.

## **C** Nonlinear Effects Including Interest Payments

FIGURE 4: MARGINAL EFFECTS OF ASSETS AND LIABILITIES ON THE PROBABILITY TO EXIT UNEMPLOYMENT BEFORE 12 MONTHS INCLUDING INTEREST PAYMENTS.



Note: This figure displays the estimated marginal effects and 95 percent confidence bands, where the four wealth groups are split into groups. The outcome variable is finding a job in the first twelve months after the beginning of unemployment. Interest payments, month and year dummies as well as individual fixed effects are added. All controls are added (age<sup>2</sup>, female, single vs. married/cohabitating, educational level, experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years, total income, wage income each year for the last three years, and hourly wage each year for the last three years).

FIGURE 5: NONLINEAR EFFECTS OF ASSETS AND LIABILITIES ON THE PROBABILITY TO BECOME RE-EMPLOYED WITHIN THE FIRST 12 MONTHS AFTER UNEMPLOYMENT BEGINS USING DUMMIES AND INCLUDING INTEREST PAYMENTS.



Note: This figure displays the estimated coefficients and 95 percent confidence bands, where the four wealth groups are split into groups and dummies for each group are used. The outcome variable is the probability of becoming re-employed in the first twelve months after the unemployment spell begins. Interest payments, month and year dummies as well as individual fixed effects are added. All controls are added (age<sup>2</sup>, female, single vs. married/cohabitating, educational level, experience, tenure in previous job, occupation, member of unemployment insurance fund, experience gained each year for the last three years, degree of unemployment each year for the last three years, total income, wage income each year for the last three years).